%====================================================================%
%                  MORIOND.TEX                                       %
% This latex file rewritten from various sources for use in the      %
% preparation of the standard proceedings Volume, latest version     %
% for the Neutrino'96 Helsinki conference proceedings                %
% by Susan Hezlet with acknowledgments to Lukas Nellen.              %
% Some changes are due to David Cassel.                              %
%====================================================================%

\documentclass[11pt]{article}
\usepackage{moriond}
\pdfoutput=1

\bibliographystyle{unsrt}    
% for BibTeX - sorted numerical labels by order of
% first citation.

% A useful Journal macro
\def\Journal#1#2#3#4{{#1} {\bf #2}, #3 (#4)}

% Some useful journal names
\def\NCA{\em Nuovo Cimento}
\def\NIM{\em Nucl. Instrum. Methods}
\def\NIMA{{\em Nucl. Instrum. Methods} A}
\def\NPB{{\em Nucl. Phys.} B}
\def\PLB{{\em Phys. Lett.}  B}
\def\PRL{\em Phys. Rev. Lett.}
\def\PRD{{\em Phys. Rev.} D}
\def\PR{\em Phys. Rev.}
\def\ZPC{{\em Z. Phys.} C}

% Some other macros used in the sample text
\def\st{\scriptstyle}
\def\sst{\scriptscriptstyle}
\def\mco{\multicolumn}
\def\epp{\epsilon^{\prime}}
\def\vep{\varepsilon}
\def\ra{\rightarrow}
\def\ppg{\pi^+\pi^-\gamma}
\def\vp{{\bf p}}
\def\ko{K^0}
\def\kb{\bar{K^0}}
\def\al{\alpha}
\def\ab{\bar{\alpha}}
\def\be{\begin{equation}}
\def\ee{\end{equation}}
\def\bea{\begin{eqnarray}}
\def\eea{\end{eqnarray}}
\def\CPbar{\hbox{{\rm CP}\hskip-1.80em{/}}}
%temp replacement due to no font
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%                                                %
%    BEGINNING OF TEXT                           %
%                                                %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%newcommand{\Photo}{\includegraphics[height=35mm]{mypicture}}
\newcommand{\met}{\mbox{$\not\!\!E_T$}}
\newcommand{\jpzm}{$J^{P}=0^{-}$}
\newcommand{\jpzp}{$J^{P}=0^{+}$}
\newcommand{\jptp}{$J^{P}=2^{+}$}
\newcommand{\Photo}{}

\begin{document}
\begin{thebibliography}{99}

\bibitem{higgs}P.W. Higgs, \Journal{{\em Phys. Lett.}}{12}{132}{1964}.

\bibitem{englert}F. Englert and R. Brout, \Journal{\PRL}{13}{321}{1964}.

\bibitem{atlas}G. Aad {\it et al.}, [ATLAS Collaboration], \Journal{\PLB}{716}{1}{2012}.

\bibitem{cms}S. Chatrchyan {\it et al.}, [CMS Collaboration], \Journal{\PLB}{716}{30}{2012}.

\bibitem{tev}T. Aaltonen {\it et al.}, [CDF and D0 Collaboration], \Journal{\PRL}{109}{071804}{2012}.

\bibitem{landau}L. Landau, \Journal{Dokl.Akad.Nauk Ser.Fiz.}{60}{207}{1948}.

\bibitem{yang}C.-N. Yang, \Journal{\PR}{77}{242}{1950}.

\bibitem{atlas_spin}G. Aad {\it et al.}, [ATLAS Collaboration], ATLAS-CONF-2013-040, (2013).

\bibitem{cms_spin}S. Chatrchyan {\it et al.}, [CMS Collaboration], CMS-PAS-HIG-13-005, (2013).

\bibitem{ellis}J. Ellis, D.S. Hwang, V. Sanz, and T. You, \Journal{{\em J. High Energy Phys.}}{1211}{134}{2012}.

\bibitem{wh}V.M. Abazov {\it et al.}, [D0 Collaboration], \href{http://arxiv.org/abs/1301.6122}{arXiv:1301.6122}, (2013), accepted by \PRD.

\bibitem{vh}V.M. Abazov {\it et al.}, [D0 Collaboration], \Journal{\PLB}{716}{285}{2012}.

\bibitem{zh}V.M. Abazov {\it et al.}, [D0 Collaboration], \href{http://arxiv.org/abs/1303.3276}{arXiv:1303.3276}, (2013), accepted by \PRD.

\bibitem{alpgen}M.L. Mangano, F. Piccinini, A.D. Polosa, M. Moretti, and R. Pittau, \Journal{{\em J. High Energy Phys.}}{0307}{001}{2003}.

\bibitem{pythia}T. Sj{\"{o}}strand, S. Mrenna, and P. Skands, \Journal{{\em J.High Energy Phys.}}{0605}{026}{2006}.

\bibitem{singletop}E.E. Boos, V.E. Bunichev, L.V. Dudko, V.I. Savrin, and V.V. Sherstnev, \Journal{{\em Phys. Atom. Nucl.}}{69}{1317}{2006}.

\bibitem{madgraph}J. Alwall, M, Herquet, F. Maltoni, O. Mattelaer, and T. Stelzer, \Journal{{\em J. High Energy Phys.}}{1106}{128}{2011}.

\end{thebibliography}

\vspace*{4cm}
\title{SPIN AND PARITY IN THE \boldmath{$WH \rightarrow \ell \nu b\bar{b}$} CHANNEL AT THE D0 EXPERIMENT}

\author{ E. JOHNSON }

\address{Department of Physics and Astronomy, Michigan State University\\ 567 Wilson Road, Rm 3220\\
East Lansing, MI 48824 USA}

\maketitle\abstracts{
We probe the spin ($J$) and parity ($P$) of the Standard Model Scalar (SMS) 
using existing searches sensitive to its production rate and kinematic 
properties. In particular, we search for the SMS decaying to a pair of $b$-quarks produced in 
association with a $W$ boson decaying leptonically. By exploiting the differences in kinematic observables, namely the 
transverse mass of the final state products ($M_{T}$), we attempt to distinguish between three 
possible $J^{P}$ hypotheses: the standard model prediction (\jpzp), a 
pseudoscalar (\jpzm), and a graviton-like particle (\jptp). 
With $9.7$~fb$^{-1}$ of data collected at the D0 experiment we show the expected sensitivity to different spin 
and parity hypotheses.
}

\section{Introduction}

Following the discovery of a boson consistent with the predicted Standard 
Model Scalar (SMS)~\cite{higgs,englert} by the ATLAS~\cite{atlas} and CMS~\cite{cms} experiments at CERN 
and the evidence for its decay into two bottom quarks at the Tevatron experiments~\cite{tev}
 it is very important to test its properties. The standard model (SM) predicts that 
the SMS will be spin zero with even parity (\jpzp). The observed 
decay to a pair of photons at the ATLAS experiment eliminates spin one as a possibility according 
to the Landau-Yang Theorem~\cite{landau,yang}. Therefore, the simplest and most 
well-motivated possibilities include the standard model prediction (\jpzp), a spin zero pseudoscalar 
(\jpzm), and a spin two particle with graviton-like couplings (\jptp). 
Although ATLAS and CMS have excluded the \jpzm\ and $2^{+}$ in 
the diphoton and four lepton final states~\cite{atlas_spin,cms_spin} they have yet to study
 the $b\bar{b}$ final state.

Searches for the SMS produced in association with a $W$ or $Z$ boson (henceforth $V$ boson) 
are sensitive to the different kinematics of the three $J^P$ hypotheses. 
This is seen most starkly in the invariant mass of the final state products~\cite{ellis}, $Vb\bar{b}$.
 The mature $VH \rightarrow Vb\bar{b}$ analyses from the D0 experiment are therefore good candidates for study. 
This paper focuses on the $WH \rightarrow \ell \nu b\bar{b}$~\cite{wh} channel specifically though the 
corresponding studies in the $VH \rightarrow \nu \nu b\bar{b}$~\cite{vh} and $ZH \rightarrow \ell \ell b\bar{b}$~\cite{zh}
 channels are underway. We employ the most recent published analysis with no modifications to the 
event selection or analysis methodologies. 

\section{Data and Simulated Samples}

This analysis uses $9.7$~fb$^{-1}$ of data collected by the D0 detector at Fermilab. Our SM 
background samples and SM signal are produced and simulated using {\sc alpgen}~\cite{alpgen}, 
{\sc pythia}~\cite{pythia}, and {\sc singletop}~\cite{singletop} or estimated from data. The \jpzm\ and \jptp\ samples 
were created using {\sc madgraph}$5$ version $1.4.8.4$~\cite{madgraph}. Within the {\sc madgraph} software 
there are several non-SM models as well as the ability to create user-defined models. We follow 
the prescription in Ellis {\it et al.}~\cite{ellis}, using the Randall-Sundrum graviton model 
for the \jptp\ sample and a user-defined model for the \jpzm\ pseudoscalar sample.
 The mass of the SMS was set to $125$~GeV, close to the 
measured value of the discovered boson. The PDF set used in the generation was {\sc cteq6l1}. 
These samples were then showered using {\sc pythia}, reconstructed, and processed through the 
full detector simulation. 

\begin{figure}[htp]
\begin{center}
\includegraphics[width=0.5\linewidth]{lnujj_mt_2ttag_all_new}
\end{center}
\caption{(color online) Transverse mass of the $\ell \nu b\bar{b}$ system for the \jpzp\, \jpzm\, and \jptp\ signals. 
These are shown after reconstruction for the two tight tag channel. The histograms are normalized to 
unit area.}
\label{fig:gen}
\end{figure}

\section{Event Selection}

We summarize the event selection for the $WH \rightarrow \ell \nu b\bar{b}$ channel. More details can 
be found in A. Abazov {\it et al.}~\cite{wh}.  We require 
one lepton ($\mu$ or $e$), large missing transverse energy (\met), and two or three jets in the final state. 
We utilize b-tagging to identify the decay products of the SMS and divide the events into 
four orthogonal tagging categories: ``one tight tag,'' ``two loose tags,'' ``two medium tags,'' and 
``two tight tags.'' A Boosted Decision Tree (BDT) is trained for each tagging category 
and jet multiplicity to separate SM signal from background. 

\section{Final Variable Distributions}

In order to differentiate between the three spin-parity assignments we use the transverse mass of the 
$\ell \nu b\bar{b}$ system defined by

\begin{equation}
M_{T}^{2} = \left( E_{T}^{W} + E_{T}^{H} \right)^{2} - \left(\vec{p}_{T}^{W} + \vec{p}_{T}^{H} \right)^{2}
\label{eq:mt}
\end{equation}

\noindent
where the transverse momentum of the $W$ boson, $\vec{p}_{T}^{W}$, is defined as 

\begin{equation}
\vec{p}_{T}^{W} = \vec{\met} + \vec{p}_{T}^{\ell}
\label{eq:pt}
\end{equation}

\noindent
To illustrate the differences between the three hypotheses for this kinematic variable, the three 
signal hypotheses are shown in Fig.~\ref{fig:gen} after reconstruction. In addition to being 
able to differentiate between the test signals (i.e. the \jpzm\ and $2^{+}$ signals), Fig.~\ref{fig:mcbkgd} shows the non-SM test signals 
peaking in a different region than either the SM signal or the other backgrounds. To ensure adequate 
background statistics when performing our statistical analysis, 
we rebin our $M_{T}$ distribution so that bins above $600$~GeV are 
included as a single overflow bin. 

\begin{figure}[htp]
\begin{center}
\includegraphics[width=0.48\linewidth]{lnujj_m_t_t_2jet_2ttag_new}
\includegraphics[width=0.48\linewidth]{WH_legend}
\end{center}
\caption{(color online) Transverse mass of the $\ell \nu b\bar{b}$ system with accompanying legend.}
\label{fig:mcbkgd}
\end{figure}

\section{Statistical Interpretation}

We use a modified frequentist (CL$_{S}$) approach with a log-likelihood ratio test statistic (LLR) for two 
hypotheses: the test hypothesis $H_{1}$ and the null hypothesis $H_{0}$. The LLR test statistic is given by

\begin{equation}
LLR = -2\ln \frac{L_{H_{1}}}{L_{H_{0}}}
\end{equation}

\noindent
where $L_{H_{x}}$ is the likelihood function for the hypothesis $H_{x}$. For a typical SMS search $H_{0}$ is the 
background-only hypothesis and $H_{1}$ is the signal plus background hypothesis. We can view our analysis through 
two different paradigms:

\begin{description}
\item[$\mathbf{H_{0} = b}$ and $\mathbf{H_{1} = s_{1}+b}$] here we decide if the data looks more like it came from a background-only 
distribution or from a test signal plus background distribution
\item[$\mathbf{H_{0} = s_{0}+b}$ and $\mathbf{H_{1} = s_{1}+b}$] here we decide if our data looks more like it came from a SM signal 
 plus background distribution or from a test signal plus background distribution
\end{description}

\noindent
 To gain insight on the separation 
significance we look at the LLR distributions populated by simulated experiments, assuming Poisson statistics,
 drawn from populations of the test and null hypotheses. 
As a preliminary result, we present the LLR distributions for the \jptp\ plus background hypothesis and background-only hypothesis
for the $WH \rightarrow \ell \nu b\bar{b}$ channel (Fig.~\ref{fig:llr} (left)) and all $VH \rightarrow Vb\bar{b}$ channels 
combined (Fig.~\ref{fig:llr} (right)). The spatial separation between the signal plus background and background-only 
LLR distributions is a good illustration of how effective the analysis is at separating the signal plus background 
and the background-only hypotheses. From these LLR distributions it is clear then that we achieve 
good separation between the signal plus background vs. background-only hypotheses.

\begin{figure}[htp]
\begin{center}
\includegraphics[width=0.48\linewidth]{lvbb_graviton_CLs_noObs}
\includegraphics[width=0.48\linewidth]{lvbb_vvbb_llbb_graviton_CLs_v2}
\end{center}
\caption{(color online) LLR distributions for the $WH \rightarrow \ell \nu b\bar{b}$ channel (left) and the combination of the $VH \rightarrow Vb\bar{b}$ 
channels (right) for the \jptp\ plus background and background-only hypotheses.}
\label{fig:llr}
\end{figure}

\section{Summary}

In summary, we presented preliminary results from the D0 experiment on the spin and parity of the newly observed SMS boson 
in the $WH \rightarrow \ell \nu b\bar{b}$ channel. While presently unable to show the observed LLR line, it is clear that 
we achieve good separation between the hypotheses. 

\section*{Acknowledgments}

We thank the staffs at Fermilab and collaborating institutions,
and acknowledge support from the
DOE and NSF (USA);
CEA and CNRS/IN2P3 (France);
MON, NRC KI and RFBR (Russia);
CNPq, FAPERJ, FAPESP and FUNDUNESP (Brazil);
DAE and DST (India);
Colciencias (Colombia);
CONACyT (Mexico);
NRF (Korea);
FOM (The Netherlands);
STFC and the Royal Society (United Kingdom);
MSMT and GACR (Czech Republic);
BMBF and DFG (Germany);
SFI (Ireland);
The Swedish Research Council (Sweden);
and
CAS and CNSF (China).

%\section*{Appendix}

% We can insert an appendix here and place equations so that they are
%given numbers such as Eq.~\ref{eq:app}.
%\be
%x = y.
%\label{eq:app}
%\ee

\section*{References}



\end{document}

%%%%%%%%%%%%%%%%%%%%%%
% End of moriond.tex  %
%%%%%%%%%%%%%%%%%%%%%%


%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 
